# Network Initialization in Low-Power Wireless Networks: A Comprehensive Study

MARJAN RADI<sup>1</sup>, BEHNAM DEZFOULI<sup>1,\*</sup>, KAMALRULNIZAM ABU BAKAR<sup>1</sup>, SHUKOR ABD RAZAK<sup>1</sup> AND MALREY LEE<sup>2</sup>

<sup>1</sup>Faculty of Computing, Universiti Teknologi Malaysia, Johor 81310, Malaysia
<sup>2</sup>Center for Advanced Image and Information Technology, School of Electronics and Information Engineering, ChonBuk National University, ChonBuk 561-756, South Korea
\*Corresponding author: dezfouli@ieee.org

The increasing growth of low-power wireless networks in real-world implementations has intensified the need to develop well-organized key network building blocks. Neighbor discovery, link quality measurement and data collection are among the fundamental building blocks of network initialization process. Over the past decade, network initialization has attracted significant attention from the research community of low-power wireless networks. Accordingly, the general concern of this paper is to survey neighbor discovery, link evaluation and collection tree construction protocols, as well as, research challenges in these research areas. Furthermore, we explore the impacts of these protocols on the functionality of different layers in the network protocol stack. In order to provide a clear view of the state-of-the-art neighbor discovery approaches, this paper also presents a classification of the existing neighbor discovery protocols. Finally, some of the important open issues in developing network initialization protocols are discussed to present new directions for further research.

Keywords: low-power wireless networks; sensor networks; network initialization; neighbor discovery;

link quality; collection tree

Received 21 November 2012; revised 2 April 2013 Handling editor: Raif Onvural

# 1. INTRODUCTION

In large-scale low-power wireless networks (e.g. wireless sensor networks), nodes are randomly deployed in a large area without any communication infrastructure. In order to initiate the network operation, nodes must configure themselves and construct the network topology after their deployment. Moreover, due to the multi-hop communication pattern of these networks, all the nodes should establish multi-hop paths toward the common network collection point (i.e. root node in the network tree) to provide a data collection service for the underlying applications. Therefore, neighbor discovery and collection tree construction are the indispensable parts of network initialization in wireless networks [1, 2].

Although neighbor discovery and collection tree protocols provide simple services, the power constraints of lowpower nodes and unpredictable behavior of wireless links introduce a lot of challenges in designing these protocols [3]. Since low-power wireless links are highly dynamic, nodes should be aware of data transmission quality of the links toward individual neighboring nodes to maintain correct network operation. Extensive empirical studies of the behavior of wireless communications have shown that link quality variations significantly influence the performance of neighbor discovery, localization, topology control, routing and medium access control (MAC) protocols [4]. However, the timevarying properties of low-power wireless links have been usually neglected in the existing communication protocols. Specifically, it is usually assumed that perfect communication (i.e. ideal packet reception rate (PRR)) can be achieved inside a specific communication range [5]. Nevertheless, due to the unpredictable behavior of low-power links, neighbor discovery and collection tree protocols may expose a complex behavior under real-world settings (e.g. network partition and routing oscillations) [6]. Since the inefficiency of these protocols can

directly influence the performance of higher layer protocols, it is necessary to identify and consider all the effects of link layer issues during the network initialization phase. Accordingly, neighbor discovery and collection tree protocols should benefit from the link quality estimation techniques to maintain the overall network performance at a preferred level. Still, the accuracy of neighborhood assessment highly depends on the probe packet size, traffic rate and environmental conditions.

Despite the static nature of low-power nodes in most of the applications, network connectivity may change frequently due to the restricted power supply, time-varying properties of wireless links, node failures, multipath effect and noise floor [7]. Therefore, in addition to the network initialization phase, neighbor discovery, link estimation and collection tree construction processes must be performed as an ongoing process in the course of network operation to identify network topology changes and update the neighborhood information of the nodes with minimum interruption in the network operation [8, 9]. Since these continuous processes require frequent network scans, channel scan frequency and beaconing rate of the nodes play an important role in providing an energyefficient and timely response against the network dynamics.

While the network initialization is a non-trivial task in wireless networks, the existing network initialization approaches have not been surveyed in detail so far. There is only one relevant survey by Galluzzi and Herman [10] that reviewed the discovery problem in wireless sensor networks without considering other critical network building blocks at the initialization phase (i.e. neighborhood assessment and collection tree formation). They briefly introduced sensor node platforms and presented some background concepts of distributed computing with a focus on the wakeup problem in neighbor discovery. However, their work cannot provide a clear classification of the previous efforts which have been done on the initialization phase of low-power wireless networks. Moreover, it does not present any information regarding the impacts of wireless propagation on the efficiency and accuracy of network initialization. Due to the lack of comprehensive studies of the initialization of low-power networks, it is difficult for researchers to obtain a coherent view of the initialization problem and find open research issues.

The aim of this study is to provide a clear view of the stateof-the-art network initialization approaches and analyze the reported observations from the previous studies. These analyses can provide useful information for the network designers to make the existing protocols as efficient as these approaches can be or develop entirely new mechanisms according to the main features of low-power wireless networks. In addition, through identifying the effects of network initialization on the performance of different layers of the network protocol stack, importance of the network initialization phase is highlighted.

The rest of this paper is organized as follows. Section 2 analyzes the main characteristics of low-power radio links and demonstrates the importance of understanding link layer issues for designing efficient network protocols. Section 3 provides an overview of the initialization of low-power wireless networks. The main challenges in supporting efficient network setup in low-power wireless networks are studied in Section 4. Section 5 investigates the key design issues in developing network initialization protocols. Section 6 clarifies the impacts of network initialization approaches on the performance of different protocols. A comprehensive taxonomy on the existing neighbor discovery techniques is presented in Section 7. Section 8 introduces some directions for further research. Finally, Section 9 concludes the paper.

# 2. AN OVERVIEW OF THE LOW-POWER WIRELESS LINKS

Various observations have motivated the research community of wireless networks to focus on the characterization of low-power radio propagation [4, 5]. The main purpose of these research studies is to find the best solutions to overcome the distractive properties of wireless communications. Link layer information can be used to provide the QoS requirements of different applications. For example, routing metrics can be derived using the transmission quality of wireless links to select the best forwarder nodes at each hop, which can reduce the probability of packet loss and the required number of retransmissions at the MAC layer to recover the lost frames. Moreover, utilizing this information for path construction reduces the cost of data transmission throughout the network, while it also improves the end-to-end network throughput and data delivery ratio [11, 12].

Over the past decade, several empirical studies of the characteristics of low-power wireless links have been carried out using various radio chips (e.g. CC1000 and CC2420) [5, 13, 14]. Based on these observations, non-ideal characteristics of lowpower wireless links (e.g. interference, irregular radio range, asymmetric communications, dynamic and unpredictable behavior) propose new issues that should be addressed in designing reliable and efficient communication protocols [15, 16]. Due to the high sensitivity of low-power radio signals to the noise, interference and multipath effect, packet delivery ratio of low-power wireless links fluctuates frequently during the network operation [17]. In order to reduce the effects of link variations on the functionality of network protocols, link quality measurement is an essential part for protocol design [4, 18]. In this context, protocol designers should have a detailed knowledge about the properties of low-power radios and wireless communications to design efficient network protocols that consider the real-world characteristics of wireless links and low-power radios.

Different studies with various network and environmental settings (e.g. packet size, traffic load, indoor and outdoor environments) have demonstrated that low-power wireless links can be classified in three individual regions: connected region, transitional region and clear region [19, 20]. As it can be seen

2



FIGURE 1. (a) PRR variations against distance [104]. (b) PRR fluctuations over time in different regions [105].

from Fig. 1a, all the nodes inside the connected region have good connectivity and a high PRR. However, the size of this region (i.e. connected region) is highly correlated with the transmission power. In contrast, clear region includes poor-quality links, which cannot be used for data transmission. Between these two regions (i.e. in the transitional region), the average data reception rate declines evenly, but different links in this region represent high variations in their transmission quality.

Figure 1b demonstrates PRR variations over time in the connected region (i.e. a receiver near the transmitter), transitional region (i.e. gray area) and clear region (i.e. a receiver on the edge of transmission range). According to this figure, wireless links in the connected region have small or no variation over time; while the transmission quality of the links in both of the transitional and clear region shows considerable variations. Environmental characteristics, multipath fading of radio signals, shadowing effect of the human activities, transmission power variations, noise floor variations and interference are the main causes of temporal variations in link quality [21–23].

Besides, the irregular radio range of low-power nodes is another important spatial property of wireless links. In this context, empirical studies have confirmed that the communication coverage of low-power nodes is highly variable in different directions, which results in unpredictable and asymmetric communication quality. This phenomenon is known as the radio irregularity and it can be caused by several factors such as variations in the transmission power, antenna gain, background noise and signal path loss [24]. Radio irregularity is studied in [25] through measuring the variations of received signal strength indicator (RSSI) and PRR over a wireless link with the changes in the angular direction of a receiver node. The achieved results from this study show that the radio range of low-power nodes is highly variable in different directions and it causes a high percentage of asymmetric links and radio interference. Extensive studies of the wireless link behavior show that the links located in the connected and disconnected regions are usually symmetric while most of the links in the transitional region are asymmetric [21, 26]. Figure 2a and b depict the variations of the RSSI values in different directions between two nodes, which are located 10 and 20 feet away from each other respectively. According to this figure, the RSSI value varies continuously by changing the angular direction of the receiver node. This observation confirms the anisotropic property of link quality in low-power wireless networks, which directly influences the functionality of different protocols (e.g. neighbor discovery, collection tree, MAC, routing, topology control and localization protocols) [25]. For example, radio irregularity can disturb the channel reservation operation of the MAC protocols that rely on the carrier sense multiple access/collision avoidance (CSMA/CA) mechanism and decrease the data frame delivery ratio at the MAC layer. Furthermore, as radio irregularity increases the percentage of asymmetric links, it may adversely affect the operation of neighbor discovery and routing protocols. For example, during the neighbor discovery process, node A may receive a beacon message from node B. Accordingly, node A will select node B as its neighboring node, while due to the link asymmetry, node B cannot hear any packet from node A and data transmission from node A to node B is subject to fail. In addition, path establishment without considering the temporal and spatial variations of link quality may result in unstable paths. As successful data delivery over low-quality paths requires a large number of retransmissions at each hop, network performance will be reduced in terms of throughput, delay, data delivery ratio and lifetime. Radio irregularity also elevates the localization errors and it makes topology maintenance troublesome. In this context, utilizing link layer information in designing topology control protocols can help to select high-quality and long-lived links in order to avoid frequent



**FIGURE 2.** (a) Variations of the RSSI in different directions, when the distance between the sender and receiver is 10 feet. (b) Variations of the RSSI in different directions when the distance between the sender and receiver is 20 feet.

topological changes. Accordingly, as neighbor discovery and collection tree construction are the fundamental parts of different network protocols, it is vital to consider wireless link characteristics in designing these techniques.

A wide range of link quality assessment techniques have been developed for wireless networks [4]. The existing protocols utilize the provided valuable link quality information by the different layers of network protocol stack to perform link quality estimation process. For instance, physical layer can provide bit level information during the reception of individual packets. Therefore, using physical layer information for link quality measurement can help to avoid spending time on the poor and borderline links. Furthermore, link layer can provide some information regarding the packet delivery statistics. Using the information provided by the link layer, the link estimation method can measure the bidirectional link quality for delivering data packets to the receiver nodes and receiving their acknowledgments at the sender nodes according to the network traffic [27]. As demonstrated in Fig. 3, the existing link quality measurement mechanisms can be generally classified in hardware-based, software-based and hybrid approaches based on the utilized information sources for assessing the data delivery performance of wireless links.

#### 2.1. Hardware-based link quality estimation

This category includes several link quality indicators (LQI), such as RSSI, LQI and signal-to-noise ratio (SNR), which can be obtained from the radio transceiver. These metrics can provide a fast estimation about high-quality links (i.e. connected region) and prevent link estimation methods from spending time on measuring the quality of marginal or poor quality links [28]. In the following subsections, the capability of individual hardware-based LQIs are investigated.

#### 2.1.1. RSSI-based link quality estimation

RSSI is a signal-based LQI that represents the observed signal strength at the receiver node during the preamble reception. The valid range of RSSI values depends on the transceiver type. Therefore, the observed RSSI values can be translated



**FIGURE 3.** Taxonomy of the existing link quality estimation techniques for low-power wireless networks.



**FIGURE 4.** RSSI variations cause by the transmission power of an interfering node [70].

into a specific range, according to the specifications of different wireless cards. Since RSSI values are computed based on the preamble bytes, this indicator does not reflect the average received signal strength during the reception of a whole packet. As a result, the RSSI metric cannot capture the effects of wireless interference during the reception of payload data. Figure 4 demonstrates the effects of wireless interference on the RSSI fluctuations with three nodes (e.g. a sender node, a receiver node and an interferen node). During this experiment, the interference



**FIGURE 5.** Correlation between RSSI and PRR in concurrent transmissions scenario under different traffic rates [70].

node continuously broadcasts data packets in the course of data transmission between the sender and receiver. According to this figure, while the RSSI values for different interference levels are almost constant, PRR decreases dramatically as the wireless interference level increases. Given that the RSSI values are recorded during the preamble reception, if the wireless interference occurs during the reception of payload data, the RSSI value still remains constant. Moreover, if the reception of a preamble fails due to wireless interference, the RSSI value cannot be recorded. Furthermore, as the preamble bytes are transmitted at a lower data rate, if the remaining bytes of the packet are transmitted at a higher data rate, no correlation is expected between the RSSI values and PRR. This is due to the fact that high data transmission rates are more prone to noise and wireless interference than low data transmission rates. Therefore, a low level of wireless interference may result in a high bit error rate (BER). The effects of network traffic rate on the relationship between RSSI metric and PRR have been shown through Fig. 5. As expected, for lower traffic rates there is a correlation between RSSI values and PRR, but under high traffic rate conditions, the RSSI values are not correlated with the measured PRR. These observations confirm that RSSI values cannot reflect the exact value of experienced interference level during data transmission. Another drawback of the RSSI metric is that the RSSI value indicates the received signal strength from the transmitter, plus external noise and experienced interference from simultaneous transmissions. Since interfering transmissions can elevate the RSSI values, using this indicator in the presence of interfering transmissions causes inaccurate measurements. The extensive evaluations on a testbed of TelosB motes in [29] confirm that utilizing RSSI as a stand-alone metric is insufficient to classify the whole link quality spectrum in a low-power wireless network. As can be seen from Fig. 6, RSSI-based link quality estimator represents similar packet delivery performance for all the links with the RSSI values more than  $-87 \, \text{dBm}$ , which is close to the sensitivity threshold of CC2420 radio (i.e. -90 dBm). However, below this threshold (i.e. lower than -87 dBm) PRR varies significantly between 0 and 100% [30]. Therefore, if the measured RSSI values are



**FIGURE 6.** Correlation between PRR and RSSI at power level 0 dBm [30].

above the sensitivity threshold, the considered links should be classified as high-quality links (i.e. links in the connected region) with a constant PRR over time (around 99%) [5, 31]. In contrast, those links with RSSI values below the sensitivity threshold should be categorized as medium or low-quality links (i.e. links in the transitional or disconnected region).

#### 2.1.2. LQI-based link quality estimation

Newly designed radios based on the IEEE 802.15.4 standard (e.g. CC2420 radio) can provide the LQI index directly from the radio module. The LQI index in these radios is computed based on the first eight symbols of the successfully received packets. However, the retrieved index by this metric depends on its implementation. For instance, in CC2420 radio, LQI indicates the chip error rate.

According to Fig. 7a, as the RSSI value increases, the LQI index is also elevated. This means that when the observed RSSI values on the wireless links are above the sensitivity threshold, the LQI metric can classify the examined links as high-quality links (e.g. links with 100% PRR). However, due to the high LQI variations of intermediate-quality links over short periods of time, it cannot accurately classify intermediate-quality links. For instance, in Fig. 7b the LQI index of a link with 0.2 PRR is similar to the calculated LQI of a link with 0.8 PRR.

As it can be seen in Fig. 8a, the transmission quality of stable links is almost constant over long periods and a single LQI reading is enough to classify these links. Nevertheless, the high variations of intermediate-quality links over short periods of time (Fig. 8b) require averaging a large number of LQI readings (e.g. at least 40 and up to 120 packets) [30, 32]. However, calculating the average value of a large number of LQI readings is in contrast with the agility of link quality estimation methods. Based on these observations, LQI can identify high-quality links, but it is incapable of providing accurate estimations regarding the intermediate-quality links.

#### 2.1.3. SNR-based link quality estimation

This metric indicates the ratio of the received signal to the noise power (the noise power includes the environmental and



FIGURE 7. (a) Correlation between LQI and RSSI [29]. (b) Correlation between PRR and LQI [39].



FIGURE 8. Variations of PRR over stable and unstable links [53].

hardware noise). Accordingly, SNR can provide better estimates of the link quality than the RSSI because the RSSI value indicates the sum of received signal power plus noise floor. Similar to the other hardware-based metrics, SNR can only differentiate between high-quality links and other links [33]. As it is shown in Fig. 9, if the calculated SNR for a link is above 20 dB, it can be classified as a high-quality link. However, links with the SNR values between 5 and 10 dB cannot be easily distinguished. Different empirical studies show that the relation between SNR and PRR highly depends on the node hardware and environmental factors such as temperature [34].

#### 2.2. Software-based link quality estimation

PRR, required number of packet (RNP) transmissions and BER metrics can be categorized as the software-based LQIs. The following subsections are dedicated to describe different aspects of these metrics.



FIGURE 9. Correlation between PRR and SNR [39].

#### 2.2.1. PRR-based link quality estimation

PRR identifies the ratio of the received packets at a receiver to the number of transmitted packets by the sender node during a predefined time window. Accuracy of this metric highly depends on the size of the selected time window for link monitoring, network traffic rate and probe packet size. With respect to the high variations of intermediate-quality links, accurate link estimations can be achieved by measuring the PRR metric over a large time window. Also, due to the low variations of highquality links, utilizing a small time window would be sufficient to yield accurate estimations.

Measuring the PRR of intermediate-quality links over a large time window can provide more accurate results; however, transmitting a large number of control packets for link measurement will reduce the agility and responsiveness of the link quality estimators against link fluctuations. Therefore, the time window adjustment should exploit a trade-off between the accuracy and agility of the link quality estimator so that it can provide a sufficient number of packet exchanges to obtain accurate link estimations in the shortest possible interval. In order to provide a stable network performance against short-term link quality variations, some protocols utilize filter-based link quality estimators (e.g. moving average, exponentially weighted



FIGURE 10. PRR variations against different packet sizes [70].

moving average and time-weighted moving average) to smooth the estimated link quality values [35, 36]. These link quality estimators use a historical control factor to control the effects of previous link quality estimations on the new estimations. Meanwhile, the utilized historical factor has a high impact on the performance of these estimators. For instance, they can provide more stable network performance as the historical factor increases. However, this factor should be properly adjusted to provide stability with a reasonable responsiveness.

In addition to the time window size, the accuracy of the PRRbased link estimator also depends on the probe packet size [37]. There is a lower probability that small-size data packets be affected by external noise and wireless interference. Thus, using small probe packets cannot result in accurate estimations. Figure 10 shows the effects of probe packet size on the PRR of three randomly selected links in a wireless testbed. According to this figure, the data transmission quality of the selected links is highly affected by the changes in the packet size.

#### 2.2.2. RNP-based link quality estimation

RNP metric estimates the RNP transmissions or retransmissions to achieve a successful packet reception. The sender node can realize the number of successfully transmitted packets through counting the number of received acknowledgment packets. Therefore, RNP-based link quality estimation technique requires an automatic repeat request mechanism at the links layer to enable the network nodes to repeat packet transmission until a successfully reception [4, 38].

Empirical studies of the temporal properties of wireless links in a test-bed of Mica motes showed that the RNP metric provides better link quality estimations than the PRR-based mechanism [21]. This is due to the fact that RNP considers the underlying packet loss distribution and RNP retransmissions to provide reliable estimations. In fact, if a successful data delivery over a link is achieved after one or more retransmissions, the PRR metric will report the same data transmission quality for that link. Whereas, RNP can reflect the inefficiency of lowquality links, which require a large number of retransmissions before successful reception of individual packets. To confirm this issue, the authors in [21] measured the ability of the PRR and RNP methods to characterize good-, intermediate- and bad-quality links over several hours. The measurements are demonstrated in Fig. 11. According to Fig. 11a and b, the PRRand RNP-based methods measure the data transmission quality of a low-quality link with an average reception rate of 48.02% and a very high-quality link with an average reception rate of 95.36% in the same manner. However, as it is shown in Fig. 11c and d utilizing the PRR-based method to indicate the data transmission performance of intermediate-quality links (i.e. with average reception rates of 86.39 and 79.86%) overestimates the actual quality of those links for data transmission. Based on these figures, although the link between node 23 and node 32 has better PRR than the link between node 23 and node 44, by using the RNP-based link quality estimation method, the second link (i.e. links 23-44) provides more efficient communication than the first one (i.e. links 23-32). The rationale behind this behavior is that short intervals of packet loss increase the RNP metric, while during these periods the average PRR may remain constantly high.

As measuring RNP metric relies on the number of received acknowledgment packets, link asymmetry imposes some problems in supporting reliable link characterization. For instance, if a node has an asymmetric link toward one of its neighboring nodes, sometimes it requires to perform a lot of retransmissions due to the low quality of its backward link, while the data transmission quality of its forward link is quite good. In this case, performing neighbor discovery in conjunction with an RNPbased link quality estimator can help to identify asymmetric links.

#### 2.2.3. BER-based link quality estimation

BER determines the ratio of corrupted bits to the total number of received bits. In order to obtain information from corrupted packets, the CRC functionality must be disabled. Furthermore, if the length bit of a packet is corrupted or several bits of a packet shifted by some bit positions, the BER will be measured wrongly. In the first case, the receiver tries to interpret a wrong number of payload bits. Although in the second case, it is possible that all the bits have been correctly received, but due to the shift of some bits the received packet is incorrectly marked corrupted. Therefore, a lot of pre-processing operations should be done on a large amount of data in order to calculate this metric. Similar to the hardware-based estimators, this metric is computed based on correctly or corrupted received packets. Therefore, loss of packets which may include a large number of erroneous bits or unsuccessful packet reception due to the interference during transmission of the preamble and header parts causes inaccurate BER values. Moreover, this metric also behaves like PRR under different traffic rates. Therefore, the measured BER values under low traffic rates cannot reflect the data transmission quality of links under high traffic rate scenarios.



FIGURE 11. PRR over time [21].

# 2.3. Hybrid link quality estimation

While each of the software-based and hardware-based link quality estimation methods can individually provide valuable information regarding the link characteristics, each method has its own advantages and disadvantages. Recent studies of the link quality characterization confirm that the combination of link quality estimation methods can cover the disadvantages of individual metrics and improve the performance of higher layer protocols [39, 40]. In fact, the main motivation behind designing hybrid link estimators is that each class of link estimation methods (i.e. software and hardware-based) can detect a particular link behavior. For instance, hardware-based link quality estimation can provide fast link qualification and present some immediate information on the channel quality during the reception of a packet. Since these methods can classify the wireless links according to the packet decoding quality, they can improve the agility of link quality estimation process through avoiding the time required to measure the poor and borderline links. However, as hardware-based link quality metrics are only measured based on the first symbols of the successfully received packets, and they do not consider the underlying packet loss distribution, these techniques usually overestimate the data transmission quality of wireless links in the presence of excessive packet losses. In contrast, softwarebased link quality estimation techniques measure the quality of links according to the successful and unsuccessful transmitted packets during a predefined time window. Therefore, they can provide more detailed information on the link quality variations [27, 41]. These differences have been shown in Fig. 12. As it can be seen, there is a period that the PRR over a link drops from 0.9 to 0.6, but the LQI did not change. This is because during this time the receiver could not receive all of the transmitted packets, but the received packets have been correctly decoded.

Hybrid link quality estimators are usually defined as a weighted function of software- and hardware-based link quality estimation metrics. The weight of each metric must be set according to the environmental conditions and performance demands of the underlying application. Different information sources (e.g. number of MAC layer retransmissions, PRR, RSSI



**FIGURE 12.** Difference between link quality characterization through software and hardware-based link quality estimations [27].

values and LQI) can be used to provide the required information of hybrid link quality estimation metrics.

#### 3. INITIALIZING LOW-POWER WIRELESS NETWORKS

Neighbor discovery in wireless networks is basically defined as identifying all the nodes in the communication range of a given node that can communicate with them directly. Upon a network deployment, all the nodes have to detect their neighboring nodes as soon as possible to construct the network topology. At the end of this stage, all the nodes have preliminary information about their surrounding neighborhood. This knowledge is required for performing distributed algorithms such as medium access mechanisms [42, 43], collection tree formation [9, 44], routing [45–47] and localization support [48]. When the network topology is established, nodes should start to construct the network data collection tree which allows data delivery from multiple source nodes toward one or more data collection points. Upon the construction of the network routing topology, upper layer protocols can start their normal operation to fulfill

the application demands. Due to the link dynamics, neighbor discovery and collection tree protocols should periodically update neighborhood and routing information in the course of network operation (i.e. continuous neighbor discovery) to maintain the desired network performance. Furthermore, through continuous neighbor discovery, the hidden nodes, which have not been identified during the network initialization due to the unreliability and asymmetric nature of wireless links can be recognized and joined to the network structure.

In order to provide an efficient network setup with respect to the resource limitations of low-power wireless networks and high dynamics of wireless links, different parameters must be considered in designing network initialization protocols. Accordingly, the key performance parameters are described as follows:

Node deployment and topology construction delay: Based on the achieved results from real-world deployments, installation of low-power nodes even in easily accessible areas may take several days or weeks [49, 50]. Accordingly, the deployment process increases the whole delay of network topology formation. Although the neighbor discovery delay can be reduced by enforcing each installed node to start the discovery process upon its installation, any attempt for neighbor discovery during the network deployment phase cannot provide complete information regarding the surrounding area of individual nodes. To avoid the unnecessary energy utilization during the deployment process, neighbor discovery mechanisms should reduce the period of idle listening and decrease the frequency of transmitting beacon messages before all the nodes are installed. However, this technique may increase the delay of network topology formation.

Energy efficiency: During the neighbor discovery phase, network nodes should increase their listening period to learn about the connectivity structure of their vicinity, while they should also raise their beaconing rate to elevate the probability of being discovered by the other nodes. However, these benefits can be achieved with the cost of more energy consumption. Moreover, although energy efficiency can be achieved through reducing the listening period of network nodes and decreasing channel probing frequency, it will increase the neighbor discovery delay. To overcome these issues, network nodes should aggressively participate in the neighbor discovery process at the network startup. If there is no activity or there is no significant change in the network connectivity, they can reduce their communications. Therefore, neighbor discovery should provide a trade-off between energy consumption and neighbor discovery latency.

The data collection algorithm should construct an optimal network routing topology to allow energy-efficient data collection and delivery from source nodes toward the network data collection point. To this aim, the constructed collection tree should be able to transmit the collected data to the collection point with minimum number of transmissions even in the presence of link quality variations or network connectivity changes.

Link quality awareness: Empirical studies of the link characterization have shown that PRR of more than half of the links in a low-power wireless network is highly variable from 10 to 90% over time and space [19, 51]. These studies confirmed that link quality is not exactly related to the distance from the sender. Accordingly, even if two receivers are located at the same distance from a sender, these receivers may experience different PRRs from a same sender node. Moreover, as illustrated in Fig. 1b wireless links with intermediate PRR values exploit high variations in their packet delivery performance over time. These observations demonstrate the effectiveness of link quality estimation methods to improve performance of network protocols. To this aim, data collection tree protocols can benefit from the provided link layer information by the employed link estimation techniques at the network initialization stage to construct a stable network data collection tree. By performing link quality estimation concurrent with neighbor discovery process, network nodes will be able to assess the data transmission quality of their incoming and outgoing links through utilized probe packets for neighbor discovery [49, 52]. Moreover, as rapid link quality fluctuations may result in frequent network connectivity changes and successive packet losses, all the nodes should update their neighborhood information periodically to avoid long periods of topology disconnections and packet loss.

*Responsiveness to the link quality variations*: In order to maintain stable network performance, the constructed data collection tree should adapt to the network connectivity changes without any interruption in the network operation. To this aim, neighbor discovery and link quality estimation protocols should provide complete neighborhood information at individual nodes under different environmental conditions. Furthermore, link quality estimators must be able to detect long-term link quality fluctuations and update the neighborhood information of network nodes. In this context, the link probing period (e.g. estimation time window), the frequency of link quality estimations and the utilized link monitoring technique play an important role to provide reliable response against link quality variations with minimum energy utilization.

*Robustness*: Collection tree protocols should be tolerant against the short-term variations of wireless links. Since route reconstruction is a costly and time-consuming process, the routing topology should remain stable in the presence of short-term link quality variations. However, if the employed link estimation mechanism finds link variations over a long time period (e.g. 10 min), it should stimulate higher layer protocols to react against the latest link quality estimations with minimum latency. Accordingly, constancy can be achieved through long-term estimations. Some of the existing link quality estimation techniques use filter-based methods such as exponential weighted moving average with a large smoothing factor to provide long-term link quality estimations [38]. Although, using long-term estimations can improve the robustness of link quality estimators, this mechanism significantly reduces the

agility of link quality estimators to react against link quality variations. For instance, if a link quality estimator measures the quality of wireless links frequently to provide a fast response against link quality fluctuations, it may consider short-term link quality variations in its estimations. However, in order to support a stable network performance, these short-term link quality variations can be ignored. Accordingly, an efficient link quality measurement technique should make a trade-off between robustness and responsiveness against link quality variations. Lin et al. [53] proposed the competence metric which is a long-term link quality estimation technique. This paper suggested a combination of this metric with a shortterm link estimation method such as ETX [11] to provide quick response against link quality changes, while supporting stable network performance through a long-term link quality characterization. Performance analysis of this approach shows that the combination of long-term and short-term link quality measurement techniques can result in selecting those wireless links that provide high data transmission quality in long-term and short-term periods.

Accuracy: One of the important goals in designing network initialization protocols is to maximize the possible number of neighboring nodes that individual nodes can discover during the network setup phase. Therefore, at the network initialization phase, all the nodes should intensively take place in the discovery process in order to maximize the probability of being discovered by the other nodes and elevate their chance to identify maximum possible number of neighboring nodes. Furthermore, deploying a link quality assessment technique during the network initialization phase can help to reflect the real behavior of wireless links and increase the accuracy of neighbor discovery. The accuracy of information provided by an employed link quality estimator is highly related to the utilized technique in the link quality estimation mechanisms for monitoring the link quality variations and conveying useful information on the quality of data transmission over different links. For instance, small-size beacon packets can be utilized to reduce the imposed signaling overhead by link measurement techniques. Most of the existing protocols (e.g. CentRoute [54], Beacon Vector Routing [55], S4 [56], Optimized Link State Routing [57], MintRoute and MultihopLQI) assume slow link quality changes over time. Accordingly, they rely on infrequent transmissions of small beacon packets to provide reliable data delivery [5, 19]. Since the data delivery performance of the wireless links are highly related to the size of data packets, using small-size beacon packets cannot yield accurate link quality estimations. Furthermore, as the traffic rate in the real-world deployment is variable during the network operation, infrequent beacon transmissions cannot reflect the impact of data rate variations on the transmission quality of wireless links. Therefore, selecting the best probe packet size and beaconing rate requires a trade-off between the accuracy and energy efficiency of the employed link assessment mechanism.

*Practicability*: Network initialization protocols should be easily employed and executed in resource-constrained networks without requiring additional mechanisms or hardware supports (i.e. specific radio chip features). In fact, if the initialization protocols are designed through certain algorithms, they may require an entirely new network protocol stack to set up the network. Thus, initialization protocols should preferably be simple enough to be compatible with existing protocol stacks. Furthermore, these mechanisms should be executed with minimum errors and network resource utilization.

# 4. CHALLENGES IN DESIGNING NETWORK INITIALIZATION PROTOCOLS

Minimizing network energy consumption is one of the primary concerns in designing network initialization protocols for lowpower wireless networks. For instance, in wireless sensor networks, all the nodes are battery powered and they are usually deployed in large areas. When a node runs out of battery, it is a costly and time-consuming process to replace its battery. Accordingly, the failed node disappears for a long time and this phenomenon has a significant effect on the functionality of different protocols that use the support of this node to accomplish their tasks. In those wireless networks without any serious concern about the energy consumption, usually neighbor discovery would not be performed as a separated phase. In these networks, upper layer protocols (e.g. AODV [58] and DSR [59]) use a simple broadcast technique during the network operation to enable the nodes to identify the whole set of their neighboring nodes [60]. However, each node should assume that its neighboring nodes can be only in either listen or transmit states at each time. This means that the neighboring nodes of a given node are always in the listen state unless they are in the transmission state. While this constant listening can eliminate the need for explicit neighbor discovery, it cannot provide energy-efficient communications. Despite the recent advances in radio frequency circuits design, the radio transceiver is still the major source of energy consumption in a low-power wireless node. Although the energy consumption in receive and transmit modes are different, the energy consumption of the receive mode is almost the same as that of idle listening mode. Similar to microcontrollers, radio chips also support lowpower sleep mode in which the energy consumption is almost negligible. In order to conserve energy, radio should be set into the sleep mode whenever it is not intended to transmit or receive. Accordingly, all of the designed MAC protocols for low-power networks use a periodic sleep/listen mechanism to provide energy-efficient communications [61-63]. Therefore, since nodes can receive a message during their active periods, two nodes can discover each other whenever both of them are active. However, during the network initialization phase, all the nodes have to participate intensively in the discovery phase to increase the chance of being discovered by the other nodes and identify a large set of neighboring nodes.

In addition, as at the initialization stage network nodes do not have any information regarding the duty cycle of their surrounding nodes, the neighbor discovery problem becomes more challenging. Hence, the main issue in developing neighbor discovery protocols for low-power wireless networks is that the radio should operate at low-duty cycles, while the neighbor discovery algorithm must ensure the reliability and agility of the discovery process under different conditions. Therefore, the employed scheduling algorithms in power constrained networks (i.e. periodic sleep/listen mechanisms) introduce a lot of challenges in providing accurate network initialization.

Since supporting global synchronization in large-scale low-power wireless networks is a non-trivial task, neighbor discovery algorithms should provide asynchronous node operation. In asynchronous neighbor discovery algorithms, each node can start the neighbor discovery process at a different time. In this situation, there is a high probability that network nodes miss each other's transmissions due to non-overlapping listen intervals.

The special characteristics of radio communications (e.g. wireless interference, heterogeneous transmission power levels, irregular radio propagation and link asymmetry) cause more problems in designing neighbor discovery and link quality measurement techniques [64]. For instance, neighbor discovery algorithms should be able to cope with the packet collisions caused due to the broadcast nature of wireless channel to provide accurate information. However, supporting collisionfree operations without enough knowledge regarding the neighboring nodes of each node is difficult. In addition, according to Fig. 13 changing the receiver direction can result in various PRRs from a sender node [6]. This observation confirms that anisotropic link quality also influences the accuracy of the neighbor discovery and link estimation protocols. Besides, link asymmetry which is the result of hardware variations and irregular radio propagation would cause more challenges in supporting accurate network initialization. For instance, in Fig. 14 node A can discover node B, C and D as its neighbors by receiving beacon packets from these nodes. Since the link between node A and B is asymmetric, whenever node A chooses node B as its next-hop neighboring node to forward its data packets, node B cannot hear the transmitted packets by node A. In this case, node A can receive the transmitted packets from node B, while node B cannot receive any packet form node A.

# 5. IMPLEMENTING NETWORK INITIALIZATION PROTOCOLS IN LOW-POWER WIRELESS NETWORKS

In order to configure network nodes and construct an efficient network infrastructure within an acceptable period, several phases should be considered in designing network initialization protocols. These steps are described in the following sections and are outlined in Fig. 15.



FIGURE 13. PRR from a central node at different directions [6].



FIGURE 14. Impact of the radio irregularity on the neighbor discovery.



FIGURE 15. Different phases of a typical network initialization protocol.

#### 5.1. Wakeup phase

In order to identify all of the possible network connections, network nodes should transmit beacon packets or listen to the channel during the discovery process. Therefore, any attempt for neighbor discovery before full network deployment results in energy waste. An alternative solution to provide energy-efficient network initialization is to trigger neighbor discovery process after full network deployment [49]. Therefore, when all the nodes are deployed, the discovery phase should be stimulated through programming a given node (e.g. the data collection point) to flood a wakeup message to the networks. This message usually includes the start time of neighbor discovery, number of probe packets for connectivity evaluation and channel probing interval. Since the accuracy of neighbor discovery is related to these information, the wakeup message should be delivered to all the nodes with high reliability within a specific time.

#### 5.2. Neighbor discovery and link evaluation phase

At the start of the neighbor discovery phase, all the nodes become active and start to broadcast beacon packets and listen to the channel to receive incoming beacon packets. In order to reduce signaling overhead of the link quality estimation process, initial link quality measurement can be combined with neighbor discovery through using the transmitted probe packets for link quality measurements.

Link quality measurement process is usually a combination of three procedures to provide efficient and accurate link quality information. Link probing is the first task that should be performed to assess the quality of data forwarding through individual nodes. The rationale behind considering this process is to generate traffic over the links and monitor link behavior over a predefined measurement period. There are three kinds of link probing techniques employed by the existing link quality measurement mechanisms: (1) active link probing, (2) passive link probing and (3) hybrid link probing.

In active link probing, each node probes the identified links toward its neighboring nodes through a broadcast [11] or a unicast [37] mechanism with a certain data rate. In the broadcast-based link probing, network nodes broadcast a certain number of probe packets to their neighboring nodes within a predefined period (i.e. estimation time window) without using acknowledgment and retransmission mechanisms. Since the number of broadcast probe packets is constant at all the nodes, each node can easily calculate the packet delivery ratio over the links toward its neighboring nodes by dividing the number of successfully received probe packets by the total number of transmitted packets. In the unicast-based link probing approach, the network nodes utilize a unicast mechanism to send the probe packets. The main advantage of unicast-based approach is that it can provide more accurate results on the quality of individual links than the broadcast-based mechanism [37, 65]. However, communication overhead of the unicast-based approach is directly related to the frequency of link probing and network density.

It has been proposed that passive link probing overcomes with the limitations of active monitoring through utilizing the network traffic to indicate the link quality [37, 66]. Accordingly, this mechanism can provide accurate results without causing additional communication overhead. Several empirical studies show that link quality estimation through application traffic is more accurate than the active link probing approach [67, 68]. As passive link monitoring uses the network traffic to estimate the link quality, this mechanism is not applicable during network initialization. However, it can be used during the network operation to update the transmission quality of wireless links [66]. In the passive link monitoring approach, all the nodes, including those nodes that do not participate in data transmission, should listen to the channel and overhear the data packets passing through their neighboring links. As the energy consumption of the receive mode is almost the same as that of idle listening mode, passive link probing cannot yield an energy-efficient link measurement. Moreover, whenever the network traffic is low, this mechanism cannot provide up-to-date information and may result in inaccurate estimations.

To make an efficient trade-off between advantages and disadvantages of active and passive link monitoring mechanisms, hybrid link probing is proposed. This link probing approach adaptively selects the active or passive link probing mechanism according to the network traffic. In the absence of application traffic, this approach exploits active link probing, while in the presence of application traffic it can use the existing traffic as the probe packets [41, 69]. By this combination, hybrid link probing approach can consider the impacts of various packet sizes, network traffic rate and environmental changes on the offered communication quality by different links.

Given the above classification, each link monitoring approach uses a specific traffic generation technique for link quality measurements. Since at the network initialization phase, there is no cross traffic in the network, all the nodes should transmit a predefined number of beacon messages to monitor the transmission quality of links during the neighbor discovery process.

The second task in the link quality measurement process is dedicated to fetch useful information regarding the data transmission quality of the examined links from the transmitted and received probe packets. To this aim, whenever a node receives a beacon message, it should fetch the necessary information (e.g. BER, RSSI and LQI) from the received message in order to calculate the transmission quality of the link over which this packet has been received.

Finally, when all the required information to perform link estimations are collected, the link evaluation process can take place through different mathematical expressions. The link quality metric can utilize only one or a combination of multiple LQIs to measure the transmission quality of wireless links [29, 39, 70].

As the network initialization phase includes a lot of broadcasts within a restricted time interval, the probability of packet collision is extremely high and it may decrease the accuracy of link quality estimations. Accordingly, the number of beacon packets and the beaconing interval of the nodes should be adjusted to reduce the probability of packet collision. Furthermore, link quality estimation should be performed frequently in the course of network operation to provide stable packet delivery performance during the network lifetime.

# 5.3. Collection tree construction phase

The main purpose of data collection tree protocols is to enable all the nodes to find at least one path toward a single or multiple network data collection points for data transmission [71, 72]. Different metrics (e.g. hop-count and link quality) can be used to construct a spanning tree that includes lowest cost paths [73–76]. Network nodes calculate the cost of data transmission over different paths toward the network data collection point according to the utilized cost function. Construction of the network collection tree is usually triggered by the network data collection point (e.g. sink node in wireless sensor networks) through flooding a control packet. Upon receiving the control packet, the receiver node updates the cost field of the received message according to the data transmission cost of the link from which this message has been received. If it is the first time that the receiver node receives a control packet over a link, it should broadcast the received packet after updating its cost field. Otherwise, if the updated cost is lower than the previously calculated cost, the receiver node rebroadcasts the new cost value. When the network routing structure is established, all the nodes are ready for transmitting their collected data toward the given data collection point. The employed collection tree protocol should periodically update the network connectivity and routing information to provide quick response against network dynamics. Most of the collection tree protocols use a fixed interval for updating the routing information of the nodes [77, 78]. Although increasing the beaconing interval of the nodes results in lower bandwidth and energy utilization, it reduces the responsiveness of the collection tree protocols and link estimation techniques to the link quality variations. Accordingly, to provide a quick response against topological changes with an efficient network resource utilization, an adaptive beaconing approach should be employed to adjust the beaconing rate according to the network condition [9]. For example, under normal conditions (e.g. as long as routing structure works properly and the estimated cost of data transmission over different paths is correct) the beaconing interval can be increased exponentially up to a specified maximum value to reduce the frequency of broadcasting beacon packets. Whenever, a node detects a significant change in the underlying routing topology (e.g. when a node joins or leaves the network, variations in the routing cost of the nodes or long-term link quality variations), the beaconing interval can be reduced to provide a fast response against the network dynamics [9].

# 6. IMPACTS OF NETWORK INITIALIZATION PROTOCOLS ON THE DIFFERENT LAYERS OF THE PROTOCOL STACK

Based on the empirical observations regarding the temporal and spatial behavior of low-power wireless links, the performance of higher layer protocols can be improved through performing an efficient network initialization which includes neighbor discovery, link quality estimation and collection tree construction processes. Accordingly, this section is dedicated to identify the impacts of network initialization on the performance of higher layer protocols.

#### 6.1. MAC layer

All the developed MAC protocols for low-power wireless networks (e.g. CSMA-based, time-division multiple-access (TDMA) based and hybrid protocols) exploit the available neighborhood information at individual nodes to perform different channel access functionalities (e.g. channel reservation and channel access scheduling) [79, 80]. For instance, MAC protocols can benefit from neighborhood information to measure the channel contention level for a given link during each time slot. In this context, they can use these information to determine the set of nodes that can be activated at the same time, while this selection reduces the number of collisions and results in high channel utilization [81]. Moreover, in contention-based medium access mechanisms, nodes can adjust their contention window size according to their neighborhood size to reduce idle listening and packet collision due to the concurrent transmissions or hidden node terminal problem [82]. The employed channel access scheduling techniques in TDMAbased MAC protocols also determine the channel access schedules for each node based on its two-hop neighborhood information so that concurrent data transmission from several nodes does not cause packet collision [83]. Furthermore, they can utilize the available neighborhood information at individual nodes to provide fair bandwidth utilization and avoid resource starvation problem.

Non-spherical pattern of link quality in low-power wireless networks also causes several problems during the channel reservation through contention-based MAC protocols. Therefore, link quality information can be used for further improvements in the functionality of these protocols. Figure 16a demonstrates the impact of asymmetric links on the channel reservation mechanism through Clear to Send (CTS) and Request to Send (RTS) handshaking. When node A receives an RTS message from node C, it responds to this message through transmitting a CTS message to node C. All the nodes that overhear the transmitted CTS message should postpone their packet transmission until node C completes its data transmission to reduce the probability of packet collision. However, due to the link asymmetry, it is possible that node E could not hear the transmitted CTS message by node A, while node A can hear packet transmission from node E. Therefore, concurrent packet transmission from node E and node C will result in collision at node A. Moreover, when CSMA is used as the medium access mechanism, the radio irregularity may cause the hidden terminal problem. As it is demonstrated through Fig. 16b, when node A transmits a packet to the node B, node C cannot hear this transmission. Since node C senses a clear channel, it starts packet transmission toward node B, but this transmission will result in packet collision at



FIGURE 16. (a) Impact of radio irregularity on the medium access control layer. (b) Radio irregularity and hidden terminal problem.

node *B*. Accordingly, the combination of neighbor discovery with link quality estimation can improve performance of the MAC protocols.

#### 6.2. Network layer

Due to the multihop communication pattern of low-power wireless networks, the main task of network layer is to select a set of optimal forwarder nodes between each node pair [84, 85]. In order to meet the performance requirements of different applications and provide efficient communications, routing protocols should be designed according to the performance demands of the underlying application [86-88]. Most of the early routing protocols in wireless networks rely on minimum hop-count routing [84, 89]. In the minimum hopcount routing approach, each node selects its highest distance neighboring node to minimize the number of hops along the paths. In order to increase the probability of successful data transmission over long hops, network nodes should increase their transmission power, which is in contrast with the energy constraints of low-power wireless networks. However, with a limited transmission power, the probability of packet loss over large distance hops is very high [36, 90]. This situation increases the RNP retransmissions to elevate the packet delivery ratio. Nevertheless, increasing the number of packet retransmissions causes higher energy consumption at individual nodes, more wireless interference and elevated data transmission delay. Although hop-count-based routing can provide minimum transmission delay and network resource utilization in wired networks, due to the special characteristics of radio communications (e.g. interference, fading, temporal and spatial link quality variations) this method is not suitable for low-power wireless networks [75]. According to the vast range of empirical studies of link characterization, a large area of a low-power wireless network belongs to the transitional region (where the PRR varies between 10 and 90%) [51, 91]. Accordingly, most of the wireless links in large-scale low-power wireless networks are asymmetric and unreliable [51]. Several works have shown that link asymmetry has a significant impact on the performance of multihop routing protocols [11, 19]. For instance, during packet transmission over asymmetric links a sender node may not receive any reply message from the receiver node for the delivered packets. In this case, the sender node assumes that the transmitted packet has been lost, and it will retransmit this packet until it receives a reply message. These unnecessary retransmissions waste network resources and decrease network performance in terms of throughput, delay, data delivery ratio and lifetime. Accordingly, by performing link quality measurement along with neighbor discovery process, routing protocols will be enabled to construct stable and high-quality paths to overcome link fluctuations [92–94].

#### 6.3. Transport layer

The main aim of transport layer protocols is to provide endto-end reliable packet delivery. Furthermore, these protocols can support QoS guarantee based on the performance demands of different applications, orderly transmission and network congestion control [95]. The bandwidth limitations and wireless interference of low-power wireless links can cause network congestion, which may affect the normal packet exchange rate in the network [96, 97]. For example, transmitting a large volume of data packets over low-quality links causes high packet loss ratio and increases the number of packet retransmissions to support the reliability demands of different applications. Accordingly, congestion control is the major concern in developing transport protocols to reduce packet loss rate. Furthermore, as nodes use a shared medium to communicate with each other, some of them may be poorly served due to the high dynamics of radio communications. In fact, link quality variations and the distance between the source and destination nodes cause unfair resource sharing

between different application flows. In order to support end-toend reliable data delivery, transport protocols can benefit from the information provided by the neighbor discovery and link quality estimation mechanisms to control the network traffic rate and mitigate network congestion.

# 6.4. Application layer

Link quality variations in combination with resource limitations of low-power wireless networks impose several challenges in supporting high-quality data delivery for different application traffic flows with various data transmission rates and performance demands. Since link quality variation is the main cause of unstable network performance, it should be considered a key factor for data rate adjustment. Fairness is another problem that emerges when multiple traffic flows exist in the network simultaneously [98, 99]. As different traffic flows pass over various links with dissimilar link quality, performance of the traffic flows that experience low channel quality may be influenced by the flows that are transmitted through high-quality links. Without considering the fairness issue in developing low-power wireless networks, unfair resource sharing of these networks may highly influence performance demands of network traffic flows. In order to overcome unfair resource sharing, the data transmission rate of the active application flows should be determined according to the quality of utilized paths by individual traffic flows. Furthermore, neighbor discovery and link quality estimation should be considered ongoing processes to identify new nodes that can provide better data transmission quality (in terms of network lifetime and throughput) than the active nodes along the established paths.

# 7. CLASSIFICATION OF THE EXISTING NEIGHBOR DISCOVERY PROTOCOLS

Generally, neighbor discovery can be performed in a centralized or a distributed manner [100]. In the centralized mechanisms, all the nodes should report their identities to a central node. After that, the central node determines the neighboring nodes of each node and informs all the nodes about their neighboring sets. Since the energy cost of this approach is highly related to the network size, it cannot fulfill the energy efficiency requirements of large-scale low-power wireless networks. In contrast, distributed neighbor discovery approaches allow the nodes to cooperate with each other to perform neighbor discovery. By this technique, the neighbor discovery cost is eventually distributed between all the nodes.

As can be seen from Fig. 17, the proposed neighbor discovery protocols can be categorized as synchronous and asynchronous approaches. This section describes some of the existing neighbor discovery mechanisms of each category. Furthermore, Table 1 provides an in-depth comparison of some of the existing neighbor discovery protocols based on the details of their discovery algorithms.



**FIGURE 17.** Taxonomy of the existing neighbor discovery protocols in low-power wireless networks.

#### 7.1. Synchronous neighbor discovery

In order to perform a synchronous neighbor discovery, all the nodes should be assigned the same neighbor discovery schedule. Therefore, network nodes transmit beacon messages according to a predefined transmission schedule. Furthermore, most of the existing protocols employ a CSMA/CA mechanism to control medium access and reduce the probability of packet collision during beacon transmissions. Moreover, every transmitted beacon message should include the ID and synchronization information of the message originator. Accordingly, upon the reception of a beacon message, the receiver node knows the exact time of the upcoming beacon messages and the active periods of the sender, while it also establishes a new link to its newly discovered neighboring node. In this category, the duration of each state (i.e. sleep, listen and transmit) and the beaconing interval of nodes have a high impact on the number of collisions and neighbor discovery latency. Consequently, the probability of transition between sleep, listen and transmit states can be determined based on the network density to provide reliable and energy-efficient neighbor discovery with minimum delay.

Slotted-based MAC protocols like S-MAC [79] and SCP-MAC [101] provide synchronous neighbor discovery in lowpower wireless networks. S-MAC tries to reduce network energy consumption through employing periodic sleep/listen scheduling at network nodes [79]. By this technique, nodes become active when at least one traffic flow exists in the network. Otherwise, they stay in the sleep mode. In S-MAC, all the neighboring nodes synchronize their sleep/listen schedules to reduce the network control overhead. Accordingly, they go to the sleep mode simultaneously and they wakeup again at the same time. In order to synchronize the sleep/listen schedules in the network, all the nodes should broadcast SYNC packets to exchange their schedules with their neighboring nodes. Therefore, each node should select a schedule and exchange it with its neighboring nodes before starting its periodic sleep/listen operation. Furthermore, it should also preserve the schedules of its identified neighboring nodes. To this aim, neighbor discovery is one of the important

Protocols	Synchronous	Power saving technique	Neighbor size-aware	Criteria	Time bounded	Network type	Link estimation	Mobility support No	
S-MAC [79]	Yes	Synchronized sleep/listen	No	Energy efficiency	No	Sensor	No		
Dozer [102]	Yes	Synchronized sleep/listen	No	Energy efficiency, reliability	No	Sensor	No	No	
ENDP [7]	Yes	Synchronized sleep/listen	No	Energy efficiency	No	Sensor	Yes	Yes	
Birthday protocol [50]	No	Periodic sleep/listen	Yes	Energy efficiency before and during initialization, delay	Yes	Ad hoc	No	No	
Disco [103]	No	Periodic sleep/listen	No	Delay, reliability	Yes	Sensor	No	Yes	
Quorum [63]	No	Periodic sleep/listen	No	Energy efficiency	No	Ad hoc	No	Yes	
Asynchronous neighbor discovery [100]	No	Periodic sleep/listen	Yes	Energy efficiency, delay	Yes	Sensor	No	No	
NoSE [49]	No	Low power listening	No	Energy efficiency before and during initialization, delay	Yes	Sensor	Yes	No	
Continuous neighbor discovery [8]	No	Periodic sleep/listen	Yes	Energy efficiency	Yes	Sensor	No	Yes	
U-Connected [60]	No	Low power listening	No	Energy efficiency, delay	Yes	Sensor	No	Yes	

**TABLE 1.** Comparison of the existing neighbor discovery protocols in low-power wireless networks.

operations in S-MAC. In this context, all the nodes exchange SYNC packets during the neighbor discovery period to perform synchronization process. During this process, each node first listens to the channel for a fixed interval to hear an SYNC packet from its neighboring nodes. If it did not receive any packet during this period, then it selects its own schedule and informs its neighboring nodes about its selected schedule by broadcasting an SYNC packet. Otherwise, if it receives an SYNC packet from one of its neighboring nodes before broadcasting its schedule, it will follow the announced schedule by its neighboring node. Network nodes on the border of two schedules (i.e. nodes that receive two different schedules from their neighbors) should follow both schedules and wakeup during the listen periods of these two schedules. In addition to the initial neighbor discovery, S-MAC also utilizes periodic neighbor discovery to avoid the case in which two nodes miss each other forever. Accordingly, all the nodes periodically perform neighbor discovery by listening to the channel for the whole synchronization period to hear the SYNC packets from their neighboring nodes. Dozer is a data-gathering framework that is designed to support low-power data collection in environmental monitoring applications [102]. This protocol architecture considers a coordinated interaction between MAC, topology control and a tree-based routing protocol to provide a low-power network protocol stack, which is suitable for data gathering in monitoring applications. Dozer supports energy-efficient data transmission from source nodes toward the network data collection point by using a TDMAbased MAC protocol. Since this protocol utilizes a tree-based network structure, all the nodes should be a part of the datagathering tree to provide the network data collection point with their collected data. Accordingly, each node should try to join to the network tree upon its activation. To this aim, network nodes should start to listen to the channel upon their activation to obtain some information regarding their neighborhood structure. Whenever a node connects to the network-spanning tree, it sends beacon packets at the start of its TDMA schedule to enable its neighboring nodes join the network tree. When a given node scans the whole length of a TDMA round, it analyzes the received beacon messages to select the best parent node. After that, it uses a simple back-off mechanism to set up its connection toward the selected parent node. Upon receiving a connection request packet, the receiver node finalizes the connection setup phase through assigning a time slot within its TDMA schedule and returns this information to the sender node. Energy-efficient neighbor discovery protocol (ENDP) is another neighbor discovery mechanism, which is

suitable for synchronized low-duty cycle MAC protocols in mobile wireless sensor network [7]. This protocol tries to reduce the energy overhead of beacon exchanges through utilizing MAC layer beacon packets for distributing node schedule information. Since the beacon packets of MAC layer protocols are periodically transmitted to maintain link synchronization, the neighbor discovery algorithm can reduce the signaling overhead of neighbor discovery and synchronization processes through piggybacking the two-hop neighborhood information of the nodes (including ID and synchronization information) in the payload of the existing MAC beacons. Furthermore, as these beacon messages are exchanged by a synchronized MAC protocol, utilizing these packets can help to avoid unnecessary idle periods and provide energy-efficient communications. At the network initialization phase, ENDP performs an initial network scan to enable the network nodes to identify their neighbors. In order to minimize the network scan duration, initial network scan will be finalized if a neighbor node with high RSSI is found, or a preferred number of neighboring nodes are identified. ENDP determines the preferred number of neighboring nodes according to the network topology and frequency of network dynamics. During the neighbor discovery process, each node exchanges the synchronization information of its parent with which the synchronization is currently maintained. This process replicates among the neighboring nodes of a node, and each node tries to receive a beacon message from a neighboring node that is not synchronized with that node. Upon receiving a beacon message, if the RSSI value of the received beacon is higher than the determined quality for high-quality links, the receiver node adds the sender ID to its neighborhood table. Moreover, each node also caches the received information from the nodes that their received RSSI values are adequate for communication. At the end of the neighbor discovery process, if the number of identified high-quality neighboring nodes is not enough, redundant links can be selected from the cached neighboring information. According to the operation of ENDP, network nodes can sleep and communicate with each other during the neighbor discovery process by using a synchronous beacon exchange mechanism at the MAC layer.

#### 7.2. Asynchronous neighbor discovery

In contrast to the synchronous algorithms, in asynchronous discovery protocols nodes have their own schedules for neighbor discovery and they transmit beacon messages at randomly selected times. In this category, a network node announces itself through periodically broadcasting a beacon message which includes its ID. Furthermore, network nodes should sample the wireless channel frequently to detect the transmitted beacon messages and establish the initial network topology. Whenever a node receives a beacon message, it can establish a new link toward the recently discovered neighboring node.

Birthday protocol [50] is a neighbor discovery algorithm that uses a probabilistic technique to support asynchronous neighbor discovery. The main aim of this protocol is to provide energyefficient network deployment, while it allows the network nodes to discover their neighboring nodes during the network initialization phase with high probability. In this protocol, each node can transit between the transmit (T), listen (L) or energysaving (S) states with different probabilities at the start of its own time slot to provide energy-efficient neighbor discovery. Whenever a node is in the transmit state, it advertises itself by broadcasting a beacon message, while during the listen state, a node continuously listens to the channel for incoming beacon messages. If a node receives a beacon message in this stage, it adds the included sender address in the message to its local neighbor table. Furthermore, a node can go to the sleep mode during the energy-saving state to prevent unnecessary energy consumption. Therefore, in this protocol all the nodes operate at Birthday-Listen-and-Transmit (BLT) and Birthday-Listen (BL) modes. In order to avoid unnecessary listen periods at the start of the neighbor discovery process, network nodes start to discover their surrounding areas in the BL mode. In this state, at each time slot, network nodes randomly choose one of the listen or sleep states with a random probability. Upon receiving the first message, the receiver node switches to the BLT mode. In the *BLT* mode, at each time slot a node can choose state S, state T or state L with different probabilities to sleep, transmit or receive a beacon message. Afterward, it can switch back to the BL mode which can randomly alternate between state S and T to save its energy for further operations. The neighbor discovery process will be terminated, whenever all the nodes return to the BL mode. Disco is another asynchronous neighbor discovery protocol, which allows the nodes to operate at low duty cycles, while they can discover each other without any global synchronization [103]. In this protocol, network nodes select a pair of prime numbers such that their inverses are approximately equivalent to the application-defined duty cycle. In addition, each node has a local counter and increments it with a fixed period which is predefined globally. Whenever the value of local counter at a node is dividable by either of the selected prime numbers, the node should change its state to the active mode for one period. During the active periods, each node can transmit beacon packets to introduce itself to its neighbors or listen to the channel for incoming beacon messages. By this mechanism, Disco guarantees that two nodes will have overlapped periods during a restricted period even if they have individual duty cycles. In Disco, the desired discovery latency or nodes duty cycles should be determined according to the requirements of the underlying application. After that, the prime numbers will be determined automatically according to the selected discovery latency or nodes' duty cycle. Figure 18 demonstrates the main operation of this protocol for two nodes. In this figure,  $C_i$  and  $C_j$  represent the preserved counters at node *i* and node *j*, respectively. Node *i* and node *j* select prime number 3 and 5, which are relatively prime and

[	x	0	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21	22	23
[	$c_i$	-	0	1	2	- 3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21	22
[	$c_j$	-	-	0	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21

FIGURE 18. An example of communication schedule under disco [103].

their inverse are approximately equal to the desired duty cycle of node *i* and node *j*. Node *i* and node *j* start to increment their counters at time x = 1 and x = 2, respectively. The black boxes in this figure indicate the active times of node *i* and node *j*. As it can be seen, at x = 7 and x = 22 both of the nodes have overlapped active periods. Therefore, these nodes can discover each other during these times. In addition to the discussed protocols, asynchronous neighbor discovery has also been handled through the Quorum protocol [63]. In this protocol, time is divided into the sequence intervals and then all the time intervals grouped to the sets of  $m^2$  adjacent intervals. In this protocol, m is a global parameter, which can be determined according to the desired node duty cycle. Furthermore, in each group the entire  $m^2$  intervals are arranged as a 2D matrix. A node randomly selects one row and one column of the entire intervals in each group and transits to the active state during the represented slots by the selected values. A node in the active state can transmit its beacon packets or listen for incoming beacon packets from its neighbors. By this mechanism, each node pair can be in the active state and discover each other at the time intervals represented in the intersection of their selected rows and columns. Borbash et al. [100] proposed another asynchronous neighbor discovery protocol. This protocol includes a distributed algorithm that allows the nodes to start neighbor discovery at different times. In this protocol, each node has its own time slotting and it can operate independently. Furthermore, in each slot a node can enter into the transmit or receive state with probability  $p_t$  and  $p_r$ , respectively. It is assumed that at the start of the neighbor discovery process most of the nodes are deployed in the listenonly mode, while the rest of the nodes are in the discovery mode. When a node in the discovery mode decides to transmit a beacon message during a time slot, it starts to transmit W copies of beacon message, where W has a fix positive value in the network. When a node in the listen-only mode receives a beacon message, it goes to the discovery mode and stays in this mode for a fixed duration of time slots S. During a receive slot, a node decodes its received message to identify the transmitter of the received message. After that, the receiver node makes an entry for the sender node in its neighbor table. However, due to the high probability of packet collision during beacon transmissions, the neighbor table of a node may not be completed at the end of the discovery period. In order to support energy-efficient operation from the beginning of the node deployment phase (i.e. before and during the network initialization) and provide a preliminary link quality estimation, a network initialization scheme called NoSE has been proposed in [49]. Since a complete network topology information cannot be achieved before full network deployment, in this protocol each installed node only listens to the channel at its wakeup intervals to detect the ongoing communications. When the network has been entirely deployed, a specific programmed node triggers the start of the neighbor discovery process by flooding a wakeup message to the network. In order to prepare all the nodes for initiating neighbor discovery, the wakeup message includes various information such as the beginning of the neighbor discovery process, its duration and further information such as the number of beacon packets and channel polling interval for link quality evaluation. The aim of this wakeup call is to synchronize all the nodes on a common time window to perform neighbor discovery and link estimation process. Accordingly, this synchronization dose not indicate that all the nodes synchronize their active periods for beacon transmission. When all the nodes receive the wakeup message, they start to perform the neighbor discovery and link assessment process within a fixed time window. During this process, all the nodes should exactly broadcast N beacon messages that include their identity. In order to accomplish the neighbor discovery process within a predetermined time, each node divides the discovery duration into N sub-slots and broadcasts a beacon at a randomly selected time during each sub-slot. Furthermore, every node should keep the number of received beacon messages and the maximum RSSI value for every single neighboring node in its neighborhood table for link quality estimation. By the end of the neighbor discovery process, all the nodes will have a list of their available neighboring nodes along with their respective data transmission quality.

The design of continuous neighbor discovery is discussed in [8]. This protocol tries to identify the hidden links that have not been discovered during the network initialization. A hidden connection in a set of connected nodes is identified as a direct communication link that can be identified between two nodes, while these nodes are not aware about this connection. The dashed links in Fig. 19 demonstrate the hidden links in a set of connected nodes. For instance, in this figure, node D has three hidden links toward node C, K and I. In this protocol, the whole task of neighbor discovery is divided between the nodes that have identified some of their immediate neighbor nodes during the network initialization phase. Accordingly, when a node that has been already joined to the network topology discovers a new neighboring node, it broadcasts an SYNC message to inform its previously identified neighbors about this new connection. Furthermore, to discover the hidden links from a node that is not connected to the network topology (e.g. node K in Fig. 19),



FIGURE 19. A set of connected nodes with hidden links.

the disconnected node should broadcast a *HELLO* packet at its wakeup times and listen to the channel for the incoming *HELLO* messages.

#### 8. FUTURE RESEARCH DIRECTIONS

During the past decade, several research works regarding the initialization of low-power wireless networks have been presented. However, there exist several open research issues which are not properly covered yet. This section introduces some of the important topics in this area which require further investigation.

Scalable, practical and simple implementations for network initialization: According to the resource constraints of lowpower wireless networks, it is necessary to exploit simple and scalable solutions to provide accurate network initialization with minimum cost. In this context, it is required to develop low-cost and simple protocols, which consider the main characteristics of low-power wireless links without requiring any particular hardware support or executing complex algorithms. For example, some radio transceivers do not provide LOI metric. Also, as the SNR metric cannot be easily obtained from the radio chip, a specific algorithm should be programmed at the software level. Therefore, it is necessary to design efficient link quality measurement techniques through combining those link quality metrics that can be easily and independently obtained from the employed wireless hardware. In addition, with respect to the deployment of low-power wireless networks in large areas, the number of nodes in an application may be in the order of hundred or thousand nodes. Therefore, network initialization algorithms should be scalable to work with different network sizes.

*Compatibility*: Most of the existing initialization protocols are designed based on specific assumptions such as a global network synchronization, specific MAC layer mechanisms and a fix network density. However, it is required that network initialization protocols be developed that can be executed without any limitation.

Providing optimal trade-off solutions between energy consumption and delay of neighbor discovery: Although the low-duty cycle operation is required to maximize network lifetime, it increases the delay of packet transmission. Accordingly, a good trade-off is desirable to design energyefficient neighbor discovery protocols while they also satisfy the delay requirements of different applications.

*Improving network collection throughput*: Due to the bandwidth constraints of low-power wireless networks, it is necessary to increase network throughput as much as possible. For this reason, it is essential to design routing cost metrics that consider wireless interference and data transmission quality of different links to construct a high-throughput network collection tree.

Robustness of network collection tree protocols: Collection tree protocols should be robust against wireless interference and sudden link quality fluctuations to provide stable network performance under dynamic topologies. For instance, high variations of intermediate-quality links cause routing oscillations and unstable network performance. To this aim, most of the existing collection tree protocols utilize shortterm link quality estimations. Since stable and unstable links may maintain a good data transmission quality over short periods, existing short-term link quality estimation techniques cannot accurately reflect the long-term variations of channel quality. Furthermore, as unstable links require frequent link quality measurements, the existing short-term link quality estimation techniques cannot provide low-energy cost operations. Accordingly, new approaches that can help to choose stable links for constructing the network collection tree should be investigated.

Continuous neighbor discovery and link quality estimation: Since topology of low-power wireless networks is highly dynamic, network nodes should update their neighborhood information periodically to improve responsiveness of upper layer protocols against network dynamics. In most of the existing protocols, initial and continuous neighbor discovery are performed in the same way. However, since neighbor discovery and link quality estimation during network operation can be considered the long-term processes, their optimization is necessary to provide efficient network resource utilization and increase network lifetime.

#### 9. CONCLUSION

Neighbor discovery and collection tree construction processes are essential to enable network protocols to perform their functionalities. Over the past decade, network initialization has received a notable attention from the research community of low-power wireless networks. Therefore, there was a need to summarize the ongoing research works on the initialization problem in low-power wireless networks. In this context, we have attempted to survey the fundamental concepts of network initialization.

The importance of understanding link layer behavior for designing network protocols have been shown through analyzing previous empirical studies of the packet delivery performance of low-power wireless links. We identified the effects of link dynamics on the performance of neighbor discovery and data collection protocols. Moreover, it is understood that achieving reliable neighbor discovery and highthroughput data collection capability depends on the efficiency of employed link quality measurement mechanism to provide a detailed link classification. In order to discuss the existing link quality estimation approaches, these protocols have been categorized into the hardware-based, software-based and hybrid techniques. We also examined the correlation between different LQIs as well as their pros and cons to evaluate the transmission quality of wireless links.

We identified the primary performance parameters that should be fulfilled by the design of network initialization protocols. Furthermore, we introduced different steps of the network initialization process which are required to satisfy the discussed performance parameters. Moreover, the main challenges in designing these protocols have been highlighted through different scenarios. In addition, the significance of network initialization has been clarified through discussing its impacts on the performance of different network protocols.

To provide a comprehensive review of the existing neighbor discovery protocols, we classified these approaches into synchronous and asynchronous protocols. The synchronous approaches require global time synchronization to maintain high synchronization level among network nodes. Therefore, due to the issues such as clock drifts, high signaling overhead and significant cost of maintaining global synchronization, these protocols may not be well suited for large-scale wireless networks. In contrast, the second group of neighbor discovery protocols does not require synchronization between nodes. Therefore, individual nodes can start the discovery process at different times. Since in these protocols nodes operate asynchronously, they should wakeup frequently to provide overlapped wakeup schedules. While the second approach does not require particular assumptions or implementation of complex algorithms (e.g. prior knowledge about the number of neighbors or global synchronization), it is less energy-efficient than the first approach.

Based on the analyzes and discussions presented in this paper, we identified some of the open research challenges in designing efficient network initialization protocols. Hence, we highlighted some of the future research directions to encourage new research for further investigation on the initialization of low-power wireless networks.

# REFERENCES

 Vasudevan, S., Adler, M. and Goeckel, D. (2010) Efficient algorithms for neighbor discovery in wireless network. *IEEE/ACM Trans. Netw.*, 21, 69–83.

- [2] England, D. and Veeravalli, B. (2007) A robust spanning tree topology for data collection and dissemination in distributed environments. *IEEE Trans. Parallel Distrib.*, 18, 608–620.
- [3] Hamida, E. and Chelius, G. (2006) Revisiting Neighbor Discovery with Interferences Consideration. Proc. 3rd ACM Int. Workshop on Performance Evaluation of Wireless Ad Hoc, Sensor and Ubiquitous Networks, Torremolinos, Malaga, Spain, October 2–6, pp. 74–81. ACM Press, New York, USA.
- [4] Baccour, N., Mottola, L. and Niga, M.Z. (2012) Radio link quality estimation in wireless sensor networks: a survey. ACM Trans. Sensor Netw., 8, Article 34, 33 pp.
- [5] Srinivasan, K., Dutta, P. and Tavakoli, A. (2010) An empirical study of low power wireless. ACM Trans. Sensor Netw., 6, Article 16, 49 pp.
- [6] UCLA/CSD-TR 02-0013 (2002) Complex Behavior at Scale: An Experimental Study of Low-Power Wireless Sensor Networks. Computer Science Department, University of California, Los Angeles (UCLA), USA.
- [7] Kohvakka, M., Suhonen, J., Kuorilehto, M., Kaseva, V., Hannikainen, M. and Hamalainen, T. (2009) Energy-efficient neighbor discovery protocol for mobile wireless sensor networks. *Ad Hoc Netw.*, 7, 24–41.
- [8] Cohen, R. and Kapchits, B. (2011) Continuous neighbor discovery in asynchronous sensor networks. *IEEE/ACM Trans. Netw. (TON)*, **19**, 69–79.
- [9] Gnawali, O., Fonseca, R., Jamieson, K., Moss, D. and Levis, P. (2009) Collection Tree Protocol. *Proc. 7th ACM Conf. Embedded Networked Sensor Systems (SenSys '09)*, Berkeley, CA, USA, November 4–6, pp. 1–14. ACM Press, New York, USA.
- [10] Galluzzi, V. and Herman, T. (2012) Survey: Discovery in wireless sensor networks. *Int. J. Distrib. Sensor Netw.*, 2012, 12 pp.
- [11] Couto, D., Aguayo, D., Bicket, J. and Morris, R. (2005) A highthroughput path metric for multi-hop wireless routing. *Wirel. Netw.*, **11**, 419–434.
- [12] Wang, J., Liu, Y., Li, M., Dong, W. and He, Y. (2011) QoF: Towards Comprehensive Path Quality Measurement in Wireless Sensor Networks. *Proc. IEEE INFOCOM*, Shanghai, China, April 10–15, pp. 775–783. IEEE, USA.
- [13] UCB/CSD-03-1270 (2003) Evaluation of Efficient Link Reliability Estimators for Low-Power Wireless Networks. EECS Department, University of California, Berkeley, CA, USA.
- [14] Srinivasan, K., Kazandjieva, M.A., Agarwal, S. and Levis, P. (2008) The β-factor: Measuring Wireless Link Burstiness. *Proc.* 6th ACM Conf. Embedded Network Sensor Systems (SenSys '08), Raleigh, NC, USA, November 5–7, pp. 29–42. ACM Press, New York, USA.
- [15] TR-0021 (2003) SCALE: A Tool for Simple Connectivity Assessment in Lossy Environment. UCLA Center for Embedded Network Sensing (CENS), LA, USA.
- [16] Srinivasan, K., Jain, M., Choi, J. and Azim, T. (2010) The κ Factor: Inferring Protocol Performance using Inter-Link Reception Correlation. *Proc. 16th Annual Int. Conf. Mobile Computing and Networking (MobiCom '10)*, Chicago, IL, USA, September 20–24, pp. 317–328. ACM Press, New York, USA.

20

- [17] Alizai, M.H., Wirtz, H., Kunz, G., Grap, B. and Wehrle, K. (2011) Efficient Online Estimation of Bursty Wireless Links. *Proc. IEEE Symp. Computers and Communications (ISCC)*, Kerkyra, Greece, June 28–July 1, pp. 191–198. IEEE, USA.
- [18] Baccour, N., Koub, A., Jam, M.B., Ros, D. and Becker, L.B. (2011) RadiaLE: a framework for designing and assessing link quality estimators in wireless sensor networks. *Ad Hoc Netw.*, 7, 1165–1185.
- [19] Woo, A., Tong, T. and Culler, D. (2003) Taming the Underlying Challenges of Reliable Multihop Routing in Sensor Networks. Proc. 1st Int. Conf. Embedded Networked Sensor Systems (SenSys03), Los Angeles, CA, USA, November 5–7, pp. 14–27. ACM Press, New York, USA.
- [20] Son, D., Krishnamachari, B. and Heidemann, J. (2006) Experimental Study of Concurrent Transmission in Wireless Sensor Networks. *Proc. 4th Int. Conf. Embedded Networked Sensor Systems (SenSys '06)*, Boulder, CO, USA, October 31– November 3, pp. 237–250. ACM Press, New York, USA.
- [21] Cerpa, A., Wong, J.L., Potkonjak, M. and Estrin, D. (2005) Temporal Properties of Low Power Wireless Links: Modeling and Implications on Multi-Hop Routing. *Proc. 6th ACM Int. Symp. Mobile Ad Hoc Networking and Computing (MobiHoc* '05), Urbana-Champaign, IL, USA, May 25–28, pp. 414–425. ACM Press, New York, USA.
- [22] Noda, C., Prabh, S., Alves, M. and Voigt, T. (2011) Quantifying the Channel Quality for Interference-Aware Wireless Sensor Networks. ACM SIGBED Rev., 8, pp. 43–48.
- [23] Boano, C., Voigt, T. and Noda, C. (2011) JamLab : Augmenting Sensornet Testbeds with Realistic and Controlled Interference Generation. *Processing of the 10th Int. Conf. Information Processing in Sensor Networks (IPSN '11)*, Chicago, IL, USA, April 12–14, pp. 175–186. IEEE, USA.
- [24] Zhou, G., He, T., Krishnamurthy, S. and Stankovic, J. (2004) Impact of Radio Irregularity on Wireless Sensor Networks. *Proc. 2nd Int. Conf. Mobile Systems, Applications, and Services* (*MobiSYS '04*), Boston, MA, USA, June 6–9, pp. 125–138. ACM Press, New York, USA.
- [25] Zhou, G., He, T. and Krishnamurthy, S. (2006) Models and solutions for radio irregularity in wireless sensor networks. ACM *Trans. Sensor Netw.*, 2, 221–262.
- [26] Sang, L., Arora, A. and Zhang, H. (2010) On link asymmetry and one-way estimation in wireless sensor networks. *ACM Trans. Sensor Netw.*, 6, 1–24.
- [27] Gnawali, O., Jamieson, K., Levis, P. and Fonseca, R. (2007) Four-Bit Wireless Link Estimation. *Proc. 6th Workshop on Hot Topics in Networks (HotNets '06)*, Atlanta, GA, USA, November 14–15. Citeseer, USA.
- [28] Zuniga, M., Irzynska, I. and Hauer, J. (2011) Link Quality Ranking: Getting the Best Out of Unreliable Links. Proc. Int. Conf. Distributed Computing in Sensor Systems and Workshops (DCOSS '11), Barcelona, Spain, June 27–29, pp. 1–8. IEEE, USA.
- [29] Rondinone, M. and Ansari, J. (2008) Designing a Reliable and Stable Link Quality Metric for Wireless Sensor Networks. Proc. Workshop on Real-World Wireless Sensor Networks (REALWSN '08), Glasgow, Scotland, March 31–April 4, pp. 6–10. ACM Press, New York, USA.

- [30] Srinivasan, K. and Levis, P. (2006) RSSI is Under Appreciated. Proc. 3th ACM Workshop on Embedded Networked Sensors (EmNets '06), Harvard University, Cambridge, MA, USA, May 30–31. IEEE, USA.
- [31] Srinivasan, K., Dutta, P., Tavakoli, A. and Levis, P. (2006) Understanding the Causes of Packet Delivery Success and Failure in Dense Wireless Sensor Networks. *Proc. 4th Int. Conf. Embedded Networked Sensor Systems (SenSys '06)*, Boulder, CO, USA, October 31–November 3, pp. 419–420. ACM Press, New York, USA.
- [32] Das, S., Pucha, H. and Papagiannaki, K. (2007) Studying Wireless Routing Link Metric Dynamics. *Proc. 7th ACM SIGCOMM Conf. Internet Measurement (IMC '07)*, San Diego, CA, USA, October 24–26, pp. 327–332. ACM Press, New York, USA.
- [33] LaI, D., Manjeshwar, A. and Herrmann, F. (2003) Measurement and Characterization of Link Quality Metrics in Energy Constrained Wireless Sensor Networks. *IEEE Global Telecommunications Conf. (GLOBECOM '03)*, San Francisco, USA, December 1–5, pp. 446–452. IEEE, USA.
- [34] Lin, S., Zhang, J., Zhou, G., Gu, L., He, T. and Stankovic, J.A. (2006) ATPC: Adaptive Transmission Power Control for Wireless Sensor Networks. *Proc. 4th Int. Conf. Embedded Networked Sensor Systems (SenSys '06)*, Boulder, CO, USA, October 31–November 3, pp. 223–236. ACM Press, New York, USA.
- [35] Renner, C., Ernst, S. and Weyer, C. (2011) Prediction Accuracy of Link-Quality Estimators, 16 pp. Lecture Notes in Computer Science 6567. Springer, Berlin.
- [36] Parissidis, G., Karaliopoulos, M., Baumann, R. and Syyropoulos, T. (2009) Routing Metrics for Wireless Mesh Networks. In S. Misra *et al.* (eds), *Guide to Wireless Mesh Networks*, pp. 199–230. Springer, New York, USA.
- [37] Zhang, H., Arora, A. and Sinha, P. (2006) Learn on the Fly: Data-Driven Link Estimation and Routing in Sensor Network Backbones. *Proc. 25th IEEE Int. Conf. Computer Communications (IEEE INFOCOM '06)*, Barcelona, Catalunya, Spain, April 23–29, pp. 1–12. IEEE, USA.
- [38] Baccour, N. and Jamaa, M.B. (2009) A Comparative Simulation Study of Link Quality Estimators in Wireless Sensor Networks. *IEEE Int. Symp. Modeling, Analysis and Simulation of Computer and Telecommunication Systems (MASCOTS '09)*, Imperial College, London, UK, September 21–23, pp. 1–10. IEEE, USA.
- [39] Voigt, T., Willig, A., Kay, R., Boano, C.A. and Zúñiga, M.A. (2010) The Triangle Metric: Fast Link Quality Estimation for Mobile Wireless Sensor Networks. *Proc. 19th Int. Conf. Computer Communications and Networks (ICCCN'10)*, Zurich, Switzerland, August 2–5, pp. 1–7. IEEE, USA.
- [40] Puccinelli, D. and Haenggi, M. (2008) Duchy: Double Cost Field Hybrid Link Estimation for Low-Power Wireless Sensor Networks. Proc. 5th Workshop on Embedded Networked Sensors (HotEmNets'08), Charlottesville, VA, USA, June 2–3. ACM Press, New York, USA.
- [41] Puccinelli, D. and Haenggi, M. (2010) Reliable data delivery in large-scale low-power sensor networks. ACM Trans. Sensor Netw., 6, 1–39.

- [42] Bachir, A., Dohler, M., Watteyne, T. and Leung, K. (2010) MAC essentials for wireless sensor networks. *IEEE Commun. Surv. Tutor.*, **12**, 222–248.
- [43] Kumar, S., Raghavan, V.S. and Deng, J. (2004) Medium access control protocols for ad hoc wireless networks: a survey. *Ad Hoc Netw.*, 4, 326–358.
- [44] Moeller, S., Sridharan, A. and Krishnamachari, B. (2010) Routing Without Routes: The Backpressure Collection Protocol. *Proc. 9th ACM/IEEE Int. Conf. Information Processing in Sensor Networks (IPSN '10)*, Stockholm, Sweden, April 12–16, pp. 279–290. ACM Press, New York, USA.
- [45] Radi, M., Dezfouli, B., Abu Bakar, K. and Lee, M. (2012) Multipath routing in wireless sensor networks: survey and research challenges. *Sensors*, **12**, 650–685.
- [46] Mueller, S., Tsang, R. and Ghosal, D. (2004) Multipath Routing in Mobile Ad Hoc Networks: Issues and Challenges. *Performance Tools and Applications to Networked Systems*, pp. 209–234. Lecture Notes in Computer Science 2965. Springer, Berlin.
- [47] Akkaya, K. and Younis, M. (2005) A survey on routing protocols for wireless sensor networks. *Ad Hoc Netw.*, 3, 325–349.
- [48] Li, M. and Baijian, Y. (2006) A Survey on Topology Issues in Wireless Sensor Network. *The 2nd Int. Conf. Wireless Networks* (*ICFN '10*), Las Vegas, NV, USA, June 26–29, pp. 376–380. Citeseer, USA.
- [49] Meier, A., Weise, M., Beutel, J. and Thiele, L. (2008) NoSE: Efficient Initialization of Wireless Sensor Networks. *Proc. 6th* ACM Conf. Embedded Networked Sensor Systems (SenSys '08), NC, USA, November 4–7, pp. 397–398. ACM Press, New York, USA.
- [50] McGlynn, M.J. and Borbash, S.A. (2001) Birthday Protocols for Low Energy Deployment and Flexible Neighbor Discovery in Ad Hoc Wireless Networks. Proc. 2nd ACM Int. Symp. Mobile Ad Hoc Networking and Computing (MobiHoc '01), CA, USA, October 4–5, pp. 137–145. ACM Press, New York, USA.
- [51] Zamalloa, M.Z.N. and Krishnamachari, B. (2007) An analysis of unreliability and asymmetry in low-power wireless links. ACM Trans. Sensor Netw., 3, Article 7, 34 pp.
- [52] Abdesslem, F., Iannone, L. and de Amorim, M. (2007) An Abstraction Layer for Neighborhood Discovery and Cross-Layer Metrics. *Proc. 16th IST Mobile and Wireless Communications Summit*, Budapest, Hungary, July 1–5, pp. 1–5. IEEE, USA.
- [53] Lin, S., Zhou, G., Whitehouse, K. and Wu, Y. (2009) Towards Stable Network Performance in Wireless Sensor Networks. *Proc. 30th IEEE Real-Time Systems Symp. (RTSS '09)*, Washington, DC, USA, December 1–4, pp. 227–237. IEEE, USA.
- [54] CENS-TR-58 (2005) Mote Herding for Tiered Wireless Sensor Networks. Center for Embedded Network Sensing, University of California, CA, USA, 7 December.
- [55] Fonseca, R., Ratnasamy, S., Zhao, J. and Ee, C. (2005) Beacon Vector Routing: Scalable Point-to-Point Routing in Wireless Sensornets. *Proc. 2nd Conf. Networked Systems Design and Implementation*, Boston, MA, USA, May 2–4, pp. 329–342. ACM Press, New York, USA.

- [56] Mao, Y., Wang, F., Qiu, L. and Lam, S. (2007) S4: Small state and small stretch routing protocol for large wireless sensor networks. *IEEE/ACM Trans.Netw.*, 18, 761–774.
- [57] Jacquet, P., Muhlethaler, P. and Clausen, T. (2001) Optimized Link State Routing Protocol for Ad Hoc Networks. *Proc. IEEE Int. Multi Topic Conf. (INMIC '01)*, Pakistan, December 28–30, pp. 62–68. IEEE, USA.
- [58] Perkins, C.E. and Royer, E M. (1999) Ad-Hoc On-Demand Distance Vector Routing. Proc. 2nd IEEE Workshop on Mobile Computing Systems and Applications (WMCSA'99), New Orleans, LA, USA, February 25–26, pp. 90–100. IEEE, USA.
- [59] Johnson, D.B., Maltz, D.A. and Broch, J. (2001) DSR: The Dynamic Source Routing Protocol for Multi-Hop Wireless Ad Hoc Networks. In Perkins C.E. (eds), *Ad Hoc Networking*, pp. 139–172. Addison-Wesley, Boston, USA.
- [60] Kandhalu, A., Lakshmanan, K. and Rajkumar, R. (2010) U-Connect: A Low-Latency Energy-Efficient Asynchronous Neighbor Discovery Protocol. Proc. 9th ACM/IEEE Int. Conf. Information Processing in Sensor Networks (IPSN '10), Stockholm, Sweden, April 12–16, pp. 350–361. ACM Press, New York, USA.
- [61] Dezfouli, B., Radi, M. and Abd Razak, S. (2010) A crosslayer approach for minimizing interference and latency of medium access in wireless sensor networks. *Int. J. Comput. Netw. Commun.*, 2, 126–142.
- [62] Dezfouli, B., Radi, M., Nematbakhsh, M.A. and Abd Razak, S. (2011) A Medium Access Control Protocol with Adaptive Parent Selection Mechanism for Large-Scale Sensor Networks. 24th Int. Conf. Advanced Information Networking and Applications Workshops, Singapore, March 22–25, pp. 402–408. IEEE, USA.
- [63] Seng, Y.-C., Hsu, C.-S. and Hsieh, T.-Y. (2003) Power-saving protocols for IEEE 802.11-based multi-hop ad hoc networks. *Comput. Netw.*, 43, 317–337.
- [64] Hamida, E. and Chelius, G. (2008) Neighbor discovery in multi-hop wireless networks: evaluation and dimensioning with interference considerations. *Discrete Math. Theor. Comput. Sci.*, 10, 87–114.
- [65] WSU-CS-DNC-TR-08-01 (2008) Unraveling the Subtleties of Link Estimation and Routing in Wireless Sensor Networks. Wayne State University, Detroit, MI, USA, pp. 473–474.
- [66] Zhang, H., Sang, L. and Arora, A. (2009) On the convergence and stability of data-driven link estimation and routing in sensor networks. ACM Trans. Auton. Adapt. Syst., 4, Article 18, 27 pp.
- [67] Zhang, H., Sang, L. and Arora, A. (2010) Comparison of datadriven link estimation methods in low-power wireless networks. *IEEE Trans. Mob. Comput.*, 9, 1634–1648.
- [68] Zhang, H., Arora, A. and Sinha, P. (2009) Link estimation and routing in sensor network backbones: beacon-based or datadriven? *IEEE Trans. Mob. Comput.*, 8, 653–667.
- [69] Kim, K.-H. and Shin, K.G. (2006) On Accurate Measurement of Link Quality in Multi-Hop Wireless Mesh Networks. Proc. 12th Annual Int. Conf. Mobile Computing and Networking (MobiCom '06), Los Angeles, CA, USA, September 24–29, pp. 38–49. ACM Press, New York, USA.
- [70] Vlavianos, A., Law, L.K., Broustis, I., Krishnamurthy, S.V., Faloutsos, M. and Kong, L. (2008) Assessing link quality in IEEE 802.11 Wireless Networks: Which is the Right Metric? 2008 IEEE 19th Int. Symp. Personal, Indoor and Mobile Radio

*Communications*, Cannes, France, September 15–18, pp. 1–6. IEEE, USA.

- [71] Radi, M., Dezfouli, B., Abd Razak, S. and Abu Bakar, K. (2010) LIEMRO: A Low-Interference Energy-Efficient Multipath Routing Protocol for Improving QoS in Event-Based Wireless Sensor Networks. *Proc. 4th Int. Conf. Sensor Technologies and Applications (SENSORCOMMG10)*, Venice/Mestre, Italy, July 18–25, pp. 151–157. IEEE/ACM.
- [72] TR-63 (2006) Hyper: A Routing Protocol to Support Mobile Users of Sensor Networks. Center for Embedded Network Sensing (CENS), University of California, LA, USA.
- [73] Borges, V.C., Curado, M. and Monteiro, E. (2011) Cross-layer routing metrics for mesh networks: current status and research directions. *Comput. Commun.*, 34, 681–703.
- [74] Javaid, N., Javaid, A. and Khan, I. (2009) Performance Study of ETX Based Wireless Routing Metrics. *Proc. 2nd IEEE Int. Conf. Computer, Control and Communication (IEEE-IC4 '09)*, Karachi, Pakistan, February 17–18, pp. 1–7. IEEE, USA.
- [75] Draves, R., Zill, B. and Padhye, J. (2004) Comparison of routing metrics for static multi-hop wireless networks. ACM SIGCOMM Comput. Commun. Rev., 34, 133–144.
- [76] Javaid, N. and Bibi, A. (2010) Interference and Bandwidth Adjusted (ETX) in Wireless Multi-hop Networks. *GLOBECOM Workshops*, Miami, FL, USA, December 6–10, pp. 1638–1643. IEEE, USA.
- [77] TEP 123 (2006) The Collection Tree Protocol (CTP). TinyOS Network Working Group, University Of California, Berkeley, CA, USA, pp. 1–6.
- [78] TR-729 (2011) The Collection Tree Protocol for the Castalia Wireless Sensor Networks Simulator. Department of Computer Science, ETH, Zurich, Switzerland.
- [79] Wei Ye, Heidemann, J. and Estrin, D. (2004) Medium access control with coordinated adaptive sleeping for wireless sensor networks. *IEEE/ACM Trans. Netw.*, 12, 493–506.
- [80] Rhee, I., Warrier, A., Aia, M., Min, J. and Sichitiu, M.L. (2008) Z-MAC: a hybrid MAC for wireless sensor networks. *IEEE/ACM Trans. Netw.*, 16, 511–524.
- [81] Bao, L. and Garcia-Luna-Aceves, J.J. (2001) A New Approach to Channel Access Scheduling for Ad Hoc Networks. Proc. 7th Annual Int. Conf. Mobile Computing and Networking (MobiCom '01), Rome, Italy, July 16–21, pp. 210–221. ACM Press, New York, USA.
- [82] Bachir, A., Barthel, D., Heusse, M. and Duda, A. (2005) Hidden Nodes Avoidance in Wireless Sensor Networks. Proc. Int. Conf. Wireless Networks, Communications and Mobile Computing, Wuhan, China, September 23–26, pp. 612–617. IEEE, USA.
- [83] Rhee, I., Warrier, A. and Min, J. (2006) DRAND: Distributed Randomized TDMA Scheduling for Wireless Ad Hoc Networks. *Proc. 7th ACM Int. Symp. Mobile Ad Hoc Networking and Computing (MobiHoc '06)*, Florence, Italy, May 22–25, pp. 190–201. ACM Press, New York, USA.
- [84] Al-Karaki, J.N. and Kamal, A.E. (2004) Routing techniques in wireless sensor networks: a survey. *IEEE Wirel. Commun.*, 11, 6–28.
- [85] Lou, W., Liu, W. and Zhang, Y. (2006) Performance Optimization Using Multipath Routing in Mobile Ad-Hoc and Wireless Sensor Networks. In Cheng, M., Li, Y. and Du,

D.Z. (eds), *Combinatorial Optimization in Communication Networks*. Kluwer Academic Publishers, Springer, New York, USA.

- [86] Reddy, T.B., Karthigeyan, I., Manoj, B. and Murthy, C.S.R. (2006) Quality of service provisioning in ad hoc wireless networks: a survey of issues and solutions. *Ad Hoc Netw.*, 4, 83–124.
- [87] Alwan, H. and Agarwal, A. (2009) A Survey on Fault Tolerant Routing Techniques in Wireless Sensor Networks. Proc. 3th Int. Conf. Sensor Technologies and Applications (Senosrcomm '09), Athens/Glyfada, Greece, June 18–23, pp. 366–371. IEEE, USA.
- [88] Tilak, S., Abu-Ghazaleh, N.B. and Heinzelman, W. (2002) A taxonomy of wireless micro-sensor network models. ACM SIGMOBILE Mob. Comput. Commun. Rev., 6, 28–36.
- [89] Tarique, M., Tepe, K.E., Adibi, S. and Erfani, S. (2009) Survey of multipath routing protocols for mobile ad hoc networks. *J. Netw. Comput. Appl.*, **32**, 1125–1143.
- [90] Yang, Y. and Wang, J. (2005) Designing Routing Metrics for Mesh Networks. Proc. IEEE Workshop on Wireless Mesh Networks (WiMesh '05), CA, USA, September 26, pp. 151–157. Citeseer, USA.
- [91] Meier, A., Rein, T., Beutel, J. and Thiele, L. (2008) Coping with Unreliable Channels: Efficient Link Estimation for Low-Power Wireless Sensor Networks. *Proc. 5th Int. Conf. Networked Sensing Systems*, Kanazawa, Japan, June 17–19, pp. 19–26. IEEE, USA.
- [92] Radi, M., Dezfouli, B., Abu Bakar, K., Abd Razak, S. and Nematbakhsh, M.A. (2011) Interference-aware multipath routing protocol for QoS improvement in event-driven wireless sensor networks. *Tsinghua Sci. Technol.*, 16, 475–490.
- [93] Koksal, C. and Balakrishnan, H. (2006) Quality-aware routing metrics for time-varying wireless mesh networks. *IEEE J. Sel. Areas Commun.*, 24, 1984–1994.
- [94] Li, S., Neelisetti, R.K., Liu, C. and Lim, A. (2010) Efficient multi-path routing protocol for wireless sensor networks. *Int. J. Wirel. Mob. Netw. (IJWMN)*, 2, 110–130.
- [95] Wang, C., Li, B., Sohraby, K., Daneshmand, M. and Hu, Y. (2006) A survey of transport protocols for wireless sensor networks. *IEEE Netw. Mag.*, 20, 34–40.
- [96] Hull, B., Jamieson, K. and Balakrishnan, H. (2004) Mitigating Congestion in Wireless Sensor Networks. Proc. 2nd ACM Conf. Embedded Networked Sensor System (SensysG04), Baltimore, MD, USA, November 3–5, pp. 134–147. ACM Press, New York, USA.
- [97] Wang, C., Li, B., Sohraby, K., Daneshmand, M. and Hu, Y. (2007) Upstream congestion control in wireless sensor networks through cross-layer optimization. *IEEE J. Sel. Areas Commun.*, *Spec. Issue CrossLayer Optimized Wirel. Multimed. Commun.*, 25, 786–795.
- [98] Ee, C.C.T. and Bajcsy, R. (2004) Congestion Control and Fairness for Many-to-One Routing in Sensor Networks. *Proc.* 2nd Int. Conf. Embedded Networked Sensor Systems, Baltimore, MD, USA, November 3–5, pp. 148–161. ACM Press, New York, USA.

- [99] Chen, S., Fang, Y., Member, S. and Xia, Y. (2007) Lexicographic maxmin fairness for data collection in wireless sensor networks. *IEEE Trans. Mob. Comput.*, 6, 762–776.
- [100] Borbash, S., Ephremides, A. and Mcglynn, M. (2007) An asynchronous neighbor discovery algorithm for wireless sensor networks. *Ad Hoc Netw.*, 5, 998–1016.
- [101] Ye, W. and Silva, F. (2006) Ultra-Low Duty Cycle MAC with Scheduled Channel Polling. Proc. 4th Int. Conf. Embedded Networked Sensor Systems (SenSys '06), Boulder, CO, USA, October 31–November 3, pp. 321–334. ACM Press, New York, USA.
- [102] Burri, N. and Rickenbach, P.V. (2007) Dozer: Ultra-Low Power Data Gathering in Sensor Networks. *Proc. 6th Int. Conf. Information Processing in Sensor Networks (IPSN '07)*, Cambridge, MA, USA, April 25–27, pp. 450–459. IEEE/ACM Press, USA.
- [103] Dutta, P. and Culler, D. (2008) Practical Asynchronous Neighbor Discovery and Rendezvous for Mobile Sensing Applications Categories and Subject Descriptors. *Proc. 6th Int. Conf. Embedded Networked Sensor Systems (SenSys '08)*, Raleigh, NC, USA, November 5–7, pp. 71–84. ACM Press, New York, USA.
- [104] Zuniga, M. and Krishnamachari, B. (2004) Analyzing the Transitional Region in Low-Power Wireless Links. Proc. 1st Annual IEEE Communications Society Conf. Sensor and Ad Hoc Communications and Networksn (SECON '04), Santa Clara, CA, USA, October 4–7, pp. 517–526. IEEE, USA.
- [105] Zhao, J. and Govindan, R. (2003) Understanding Packet Delivery Performance in Dense Wireless Sensor Networks. *Proc. 1st Int. Conf. Embedded Networked Sensor Systems* (SenSys '03), Los Angeles, CA, USA, November 5–7, pp. 1–13. ACM Press, New York, USA.